Author's response to reviews

Title: Fast food diet with CCl4 micro-dose induced hepatic-fibrosis - A Novel animal model

Authors:

Tarak K Chheda (tarak.chheda@connexios.com)
Pratibha Shivakumar (pratibha.s@connexios.com)
Satish Sadasivan (satish.s@connexios.com)
Harish Chanderasekharan (harish.c@connexios.com)
Yogananda Moolemath (moolemath@connexios.com)
Anup M Oommen (anup.oommen@connexios.com)
Jagannath R Madanahalli (m.r.jagannath@connexios.com)
Venkataranganna V Marikunte (m.v.venkataranganna@connexios.com)

Version: 5 Date: 29 April 2014

Author's response to reviews: see over
To,
Editorial Board of BMC Gastroenterology

Dear Editor

Thank you for consideration of our Manuscript entitled “Fast food diet with CCl4 micro-dose induced hepatic-fibrosis –A Novel animal model” (6157822321125312) as an original article in BMC Gastroenterology. Please find the revised manuscript and response to the Editorial requirements

Thank you for the consideration.

Yours sincerely,

Dr.Venkataranganna

Editorial requirements:

1) Please give details in the methods section of where the primers they used were obtained.

Ans: Source and sequence of primers have been incorporated under Methods section

2) Please use initials in your Authors Contribution statement.

Ans: Necessary changes are incorporated